Magnetic resonance enterography compared with ultrasonography in newly diagnosed and relapsing Crohn's disease patients: the METRIC diagnostic accuracy study

Stuart A Taylor, 1* Sue Mallett, 2 Gauraang Bhatnagar, 1 Stephen Morris, 3 Laura Quinn, 2 Florian Tomini, 3 Anne Miles, 4 Rachel Baldwin-Cleland, 5 Stuart Bloom, 6 Arun Gupta, 5 Peter John Hamlin, 7 Ailsa L Hart, 8 Antony Higginson, 9 Ilan Jacobs, 10 Sara McCartney, 6 Charles D Murray, 11 Andrew AO Plumb, 1 Richard C Pollok, 12 Manuel Rodriguez-Justo, 13 Zainib Shabir, 14 Andrew Slater, 15 Damian Tolan, 16 Simon Travis, 17 Alastair Windsor, 18 Peter Wylie, 19 Ian Zealley 20 and Steve Halligan 1

¹Centre for Medical Imaging, University College London, London, UK ²Institute of Applied Health Research, National Institute for Health Research Birmingham Biomedical Research Centre, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK

³Applied Health Research, University College London, London, UK

⁴Department of Psychological Sciences, Birkbeck, University of London, London, UK

⁵Intestinal Imaging Centre, St Mark's Hospital, London North West University Healthcare NHS Trust, Harrow, UK

⁶Department of Gastroenterology, University College Hospital, London, UK

⁷Department of Gastroenterology, St James's University Hospital, Leeds Teaching Hospitals NHS Trust, Leeds, UK

⁸Inflammatory Bowel Disease Unit, St Mark's Hospital, London North West University Healthcare NHS Trust, Harrow, UK

⁹Department of Radiology, Portsmouth Hospitals NHS Trust, Portsmouth, UK

¹⁰Independent patient representative, c/o Centre for Medical Imaging, University College London, London, UK

¹¹Department of Gastroenterology and Endoscopy, Royal Free London NHS Foundation Trust, London, UK

¹²Department of Gastroenterology, St George's Hospital, London, UK

¹³Department of Histopathology, University College Hospital, London, UK

¹⁴Comprehensive Clinical Trials Unit, Institute of Clinical Trials and Methodology, University College London, London, UK

- ¹⁵Oxford University Hospitals NHS Foundation Trust, Oxford, UK
- ¹⁶Department of Radiology, St James's University Hospital, Leeds Teaching Hospitals NHS Trust, Leeds, UK
- ¹⁷Translational Gastroenterology Unit, Oxford University Hospitals NHS Foundation Trust, Oxford, UK
- ¹⁸Department of Surgery, University College Hospital, London, UK
- ¹⁹Department of Radiology, Royal Free London NHS Foundation Trust, London, UK
- ²⁰Department of Radiology, Ninewells Hospital, Dundee, UK

Declared competing interests of authors: Stuart A Taylor reports personal fees from Robarts Clinical Trials Inc. (London, ON, Canada) outside the submitted work. Simon Travis reports receiving fees for consultancy work and/or speaking engagements from the following: AbbVie Inc. (Chicago, IL, USA), Centocor Inc. (Horsham, PA, USA), Schering-Plough (Kenilworth, NJ, USA), Bristol-Myers Squibb (New York, NY, USA), Chemocentryx Inc. (Mountain View, CA, USA), Cosmo Pharmaceuticals (Dublin, Ireland), Elan Pharma Inc. (Dublin, Ireland), Genentech Inc. (San Francisco, CA, USA), Giuliani SpA (Milan, Italy), Merck & Co. Inc. (Kenilworth, NJ, USA), Takeda UK Ltd (Woodburn Green, Buckinghamshire, UK), Otsuka Pharmaceuticals (Tokyo, Japan), PDL BioPharma (Nevada, NV, USA), Pfizer Inc. (San Francisco, CA, USA), Shire Pharmaceuticals UK (St Helier, Jersey), Glenmark Pharmaceuticals (Maharashtra, India), Synthon Biopharmaceuticals (Nijmegen, the Netherlands), NPS Pharmaceuticals (Bedminster, NJ, USA), Eli Lilly and Company (Indiana, IN, USA), Warner Chilcott Ltd, Proximagen Group Ltd (London, UK), VHsquared Ltd (Cambridge, UK), Topivert Pharma Ltd (London, UK), Ferring Pharmaceuticals (Saint-Prex, Switzerland), Celgene Corporation (Summit, NJ, USA), GlaxoSmithKline plc (Brentford, UK), Amgen Inc. (Thousand Oaks, CA, USA), Biogen Inc. (Cambridge, MA, USA), Enterome SA (Paris, France), Immunocore Ltd (Oxford, UK), Immunometabolism/Third Rock Ventures (Boston, MA, USA), Bioclinica Inc. (Newtown, PA, USA), Boehringer Ingelheim GmBH (Ingelheim am Rhein, Germany), Gilead Sciences Inc. (Foster City, CA, USA), Grunenthal Ltd (Aachen, Germany), Janssen Pharmaceutica (Beerse, Belgium), Novartis AG (Basel, Switzerland), Receptos Inc. (San Diego, CA, USA), Pharm-Olam International UK Ltd (Bracknell, UK), Sigmoid Pharma (Dublin, Ireland), Theravance Biopharma Inc. (Dublin, Ireland), Given Imaging Ltd (Yokneam Illit, Israel), UCB Pharma SA (Brussels, Belgium), Tillotts Pharma AG (Rheinfelden, Switzerland), Sanofi Aventis SA (Paris, France), Vifor Pharma (St Gallen, Switzerland), Abbott Laboratories Ltd (Chicago, IL, USA) and Procter and Gamble Ltd (Cincinnati, OH, USA). Simon Travis reports directorships of charities IBD2020 (Barnet, UK; UK 09762150), Cure Crohn's Colitis (Sydney, Australia; ABN 85 154 588 717) and the Truelove Foundation (London, UK; UK 11056711). Simon Travis also reports receiving fees from the following for expert testimony work and/or royalties: Santarus Inc. (San Diego, CA, USA), Cosmo Technologies Ltd (Dublin, Ireland), Tillotts Pharma AG, Wiley-Blackwell Inc. (Hoboken, NJ, USA), Elsevier Ltd (Amsterdam, the Netherlands) and Oxford University Press (Oxford, UK). Simon Travis has received research grants from the following: AbbVie Inc., the International Organization for the Study of Inflammatory Bowel Disease, Eli Lilly and Company, UCB Inc. (Brussels, Belgium), Vifor Pharma, Norman Collisson Foundation (Bicester, UK), Ferring Pharmaceuticals, Schering-Plough, Merck Sharpe & Dohme Corp. (Kenilworth, NJ, USA), Procter and Gamble Ltd, Warner Chilcott Ltd, Abbott Laboratories Ltd, PDL BioPharma Inc. (Incline Village, NV, USA), Takeda UK Ltd and the International Consortium for Health Care Outcomes Measurement. Ailsa Hart reports personal fees from AbbVie Inc., Atlantic Healthcare Ltd (Saffron Walden, UK), Bristol-Myers Squibb, Celltrion Inc. (Incheon, South Korea), Dr Falk Pharma UK Ltd (Bourne End, UK), Ferring Pharmaceuticals, Janssen Pharmaceuticals, Merck Sharpe & Dohme Corp., Napp Pharmaceuticals Ltd (Cambridge, UK), Pfizer Inc., Pharmacosmos A/S (Holbæk, Denmark), Shire Pharmaceuticals UK and Takeda UK Ltd, and non-financial support from Genentech Inc. Alastair Windsor reports personal fees from Takeda, grants from Allergan Inc. (Dublin, Ireland), personal fees from Allergan, personal fees from Cook Medical Inc. (Bloomington, IN, USA) and grants and personal fees from Bard Ltd (Crawley, UK) outside the submitted work. Andrew Plumb reports grants from the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) programme outside the submitted work, grants from the NIHR Fellowships programme during the conduct of the study and honoraria for educational lectures delivered at events

^{*}Corresponding author stuart.taylor@ucl.ac.uk

arranged by Acelity Inc. (Crawley, UK), Actavis Pharma Inc. (Parsippany-Troy Hills, NJ, USA), Dr Falk Pharma UK Ltd, Janssen-Cilag Ltd (High Wycombe, UK) and Takeda UK Ltd on the subject of inflammatory bowel disease. Ilan Jacobs reports share ownership in General Electric Company (Boston, MA, USA), which manufacturers and sells magnetic resonance imaging equipment. Charles D Murray reports personal fees from AbbVie Inc., Merck Sharpe & Dohme Corp. and Janssen Pharmaceutica outside the submitted work. Antony Higginson reports personal fees from Toshiba Corporation (Tokyo, Japan) outside the submitted work. Steve Halligan reports non-financial support from iCAD Inc. (Nashua, NH, USA) outside the submitted work, and sat on the HTA commissioning board (2008–14). Stephen Morris reports Health Services and Delivery Research (HSDR) Board membership (2014–18), HSDR Evidence Synthesis Sub Board membership (2016), HTA Commissioning Board membership (2009–13) and Public Health Research Board membership (2011–17).

Published August 2019 DOI: 10.3310/hta23420

Plain English summary

The METRIC diagnostic accuracy study

Health Technology Assessment 2019; Vol. 23: No. 42 DOI: 10.3310/hta23420

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

rohn's disease is a waxing and waning lifelong inflammatory condition that affects the colon (large bowel) and small bowel. Treatment relies on accurately determining disease extent and underlying inflammation. Colonoscopy is very good for examining the colon, but it is invasive and, at best, can only visualise a few centimetres of the small bowel, so radiological imaging is very important. Magnetic resonance enterography (a type of magnetic resonance imaging scan) and ultrasonography are both radiological tests commonly performed in the NHS, and it is unclear which method is better. We performed a study to compare the accuracy of magnetic resonance enterography and ultrasonography for determining the extent of Crohn's disease in the bowel of participants newly diagnosed and in those participants with established Crohn's disease but with suspected deterioration. We also investigated how often radiologists agree with each other during test interpretation, the participant experience of undergoing the tests and their cost-effectiveness. We compared the tests in 284 participants (133 newly diagnosed and 151 with suspected deterioration). We found that both tests were accurate for detecting the presence (97% for magnetic resonance enterography and 92% for ultrasonography) and location (80% for magnetic resonance enterography and 70% for ultrasonography) of disease in the small bowel, but magnetic resonance enterography was better than ultrasonography for both (correctly classifying disease extent in 107 more participants for every 1000 participants with Crohn's disease). Magnetic resonance enterography was similarly better than ultrasonography at determining if the bowel was inflamed. The results were similar in newly diagnosed participants and those participants with suspected deterioration. Agreement between radiologists interpreting the same images was, at best, moderate for both tests. A total of 88% of participants tolerated magnetic resonance enterography well or fairly well, which was less than the percentage (99%) of participants who tolerated ultrasonography well or fairly well. Both tests had a similar effect on the treatment decisions made by doctors. Both tests were also similar in their value for money for the NHS.

HTA/HTA TAR

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.819

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 11/23/01. The contractual start date was in September 2013. The draft report began editorial review in May 2018 and was accepted for publication in December 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Taylor et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk